Author's response to reviews

Title: Unusual presentation of Basilar artery stroke secondary to patent foramen ovale: case report

Details of authors Author 1 (Corresponding author) Name Dr Abdul Salam MRCP(UK) Institution Worcester royal hospital Address Worcester, WR5 1DD Phone 07929766116 Fax 01905733192 email asalam@doctors.org.uk Authors title Specialist registrar geriatric/general medicine Author 2 Name Dr Phil Sanmuganathan MD,FRCP Institution Worcester royal hospital, Worcester address WR5 1DD Phone 01905733146 Fax 01905733192 email phil.sanmuganathan@worcesacute.wmids.nhs.uk Authors title Consultant stroke physician Author 3 Name Dr Chris Pycock FRCP Institution Worcester royal hospital, Worcester WR5 1DD Phone 01905763333 ext 33256 fax 01905760361 email chris.pycock@worcesacute.wmids.nhs.uk Authors title Consultant physician

Authors:

Abdul Salam Specialist Registrar (asalam@doctors.org.uk)
Phil Sanmuganathan Consultant Stroke Physician (phil.sanmuganathan@worcesacute.wmids.nhs.uk)
Chris Pycock Consultant Physician (Chris.Pycock@worcesacute.wmids.nhs.uk)

Version: 3 Date: 21 May 2007

Author's response to reviews: see over
To
Clare Shaw
Assistant Editor
Biomed Central

Dear Clare,

I am hereby submitting the revised version of manuscript along with other necessary changes as suggested by you.
Many thanks

Yours sincerely

Dr A Salam
Specialist Registrar
Good Hope Hospital